Author's response to reviews

Title: Qualitative Approach to Medical School Applicant Admission System: a Case Study

Authors:

Sara S Katz (sara10@012.net.il)
Shlomo Sh Vinker (shlomovi@clalit.org.il)

Version: 2
Date: 27 June 2014

Author's response to reviews: see over
Qualitative Approach to Medical School Applicant Admission

System: a Case Study

Sara Katz\(^1\) and Shlomo Vinker\(^2\)

\(^1\)Sara Katz Ph.D, lecturer and researcher in the Education Dept.,
Head of the Research Dept. and of English for Academic Purposes Dept.,
Sha'an an Academic College. 7 Hayam Hatichon St. Kiriyat Shmuel, P.O.B. 906,
Haifa 2640007, Israel.
Telephone: 972-4-8780000; e-mail: sara10@012.net.il

She is an educational psychology researcher. Her main is in self-efficacy self-regulation, instruction and assessment. She is the author of three books and has published and presented papers on these subjects.

\(^2\)Shlomo Vinker MD, MHA is an associate professor in family medicine at the department of Family Medicine, Sackler School of Medicine, Tel Aviv University, Tel Aviv. shlomovi@clalit.org.il

*Corresponding author: Sara Katz.
Figure 1: Group profile of applicants who were accepted and began studies.